Response to reviewers’ comments (Minor revision)

1. Do not use the abbreviations when they first appear; for example, non-ceruloplasmin copper (NCC), etc.
   We have removed the abbreviations when they first appear, only when mentioned subsequently abbreviations are used

2. Please mention about the induction of metallothionein by zinc.
   We have added a note on mechanism of Zn in inducing metallothionein in the manuscript. The added lines are highlighted in the manuscript

   Thanks for the suggestion. We have added this reference in the manuscript. The added segment is highlighted